Reviewer #1:

**Scientific Quality:** Grade A (Excellent)

**Language Quality:** Grade A (Priority publishing)

**Conclusion:** Accept (General priority)

**Specific Comments to Authors:**

The manuscript is well written. Very extensive literature at the cellular level about prognostication on Gastric cancer, where some related genes described were just newly studied while some described or related to other gastrointestinal organs and not in gastric cancer yet. It opens a new area about gastric cancer to research on.

Dear editor,

Thank you for the opportunity to read and review this manuscript of “A predictive model using four ferroptosis-related genes accurately predicts gastric cancer prognosis” in World Journal of Gastrointestinal Oncology. Our study developed a predictive model based on four FRGs (NOX4, MTCH1, GABARAPL2, and SLC2A3) that can accurately predict the prognosis of GC patients and the efficacy of immunotherapy in this population. We look forward to hearing from you at your earliest convenience. Thanks very much for your attention to our paper.